Author's response to reviews

Title: The importance of comorbidity in analysing patient costs in Swedish primary care

Authors:

Sven G Engstrom (sven.engstrom@lio.se)
Lennart Carlsson (lennart.carlsson@chello.se)
Carl-Johan Ostgren (carl.johan.ostgren@lio.se)
Gunnar H Nilsson (gunnar.nilsson@nlpo.sll.se)
Lars A Borgquist (larbo@ihs.liu.se)

Version: 3 Date: 1 February 2006

Author's response to reviews: see over
The final editing of the manuscript (The importance …) have included
- changing headlines in Abstract to Background and Methods including deleting old
  headings and slight rephrasing of the text accordingly
- Changing headings throughout the manuscript so there are no longer any capitals
- Changing the first headline to Background
- Changing the next headline to Methods
- Changing one headline to Acknowledgement (without a ’s’)
- Deleting all indications in the text where to place tables.